Author's response to reviews

Title: Patient satisfaction of primary care for musculoskeletal diseases: A comparison between Neural Therapy and conventional medicine

Authors:

Joelle Mermod (jomermod@students.unibe.ch)
Lorenz Fischer (fischer.lori@bluewin.ch)
Lukas Staub (lukas.staub@memcenter.unibe.ch)
Andre Busato (andre.busato@memcenter.unibe.ch)

Version: 8 Date: 4 June 2008

Author's response to reviews:

Dear Sirs

According to the suggestions of the BMC Editorial Production Team the following changes of the manuscript were performed:

We removed the line “*These authors contributed equally to this work” and the respective superscripts from the author list.

We added the sentence “All authors read and approved the final manuscript” to section authors contribution.

We checked the endnote library according to http://www.biomedcentral.com/info/ifora/endnote/#technicalhelp and reformatted the reference list in the article.

We are, despite several trials, not able to match the number of references in the article with the number of references extracted by the BMC document system.

We now submit a new version of the article.

With best regards
A. Busato